Author's response to reviews

Title: A randomized controlled trial to investigate the influence of low dose radiotherapy on immune stimulatory effects in liver metastases of colorectal cancer

Authors:

Christoph Reissfelder (christoph.reissfelder@med.uni-heidelberg.de)
Carmen Timke (c.timke@dkfz.de)
Hubertus Schmitz-Winnenthal (hubertus.schmitz-winnenthal@med.uni-heidelberg.de)
Nuh N Rahbari (nuh rahbari@med.uni-heidelberg.de)
Moritz Koch (moritz.koch@med.uni-heidelberg.de)
Felix Klug (f.klug@dkfz.de)
Falk Roeder (f.roeder@dkfz.de)
Lutz Edler (l.edler@dkfz.de)
Jürgen Debus (juergen.debus@med.uni-heidelberg.de)
Markus W Buechler (markus.buechler@med.uni-heidelberg.de)
Philipp Beckhove (p.beckhove@dkfz.de)
Peter E Huber (p.huber@dkfz.de)
Juergen Weitz (juergen.weitz@med.uni-heidelberg.de)

Version: 6 Date: 13 September 2011

Author's response to reviews:

All the requested changes are done (see below):

Formatting changes to be made:
----------------------------------------------

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

Authors' contributions: Please use initials instead of full name.

References: Please make sure references are cited in numerical order.

Additional files: Please remove additional file.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Abstract: Please make sure trial registration is not a bold heading.
Structure: Change literature to reference list.

Authors' contributions: Please place before the reference list.

Authors' contributions: please ensure this is formatted into a paragraph rather than a list or in bullet points.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.